Author's response to reviews

Title: A Bleeding Kiss: Intramural haematoma secondary to balloon angioplasty

Authors:

Cheerag Shirodaria (cshirodaria@doctors.org.uk)
William J van Gaal (william@vangaal.org)
Adrian P Banning (adrian.banning@orh.nhs.uk)

Version: 3 Date: 5 June 2007

Author's response to reviews:

Please find a revised version of the manuscript enclosed.

1. Highlighting/Tracking - Please remove any highlighting/tracking from your manuscript text.
   Ths has been done

2. Running head - Please remove the running head from the title page.
   This has been done.

3. Qualifications - Please remove the authors' qualifications (e.g. PhD / MD) from the manuscript and
   submissions system.
   Qualifications have been removed.

4. References - References should be cited in the text using consecutive numbers in square brackets
   starting from [1] etc. Please ensure the type of bracket used is consistent. Please do not use superscript
   numbers.
   This has been done.

5. Figure legends Please move these to a position after the references
   This has been done.

6. Authors' contributions - Please indicate that all authors read and approved the final manuscript.
   This has been done.